Author's response to reviews

Title: Staphylococcus aureus bacteriuria as a prognosticator for outcome of Staphylococcus aureus bacteremia: a case-control study

Authors:

Shingo Chihara (shingo_chihara@rush.edu)
Kyle J Popovich (kyle_popovich@rush.edu)
Robert A Weinstein (rweinste@rush.edu)
Bala Hota (bhota@rush.edu)

Version: 7 Date: 27 July 2010

Author's response to reviews:

I made 3 revisions since the last one.

1. CCH in the methods section of the abstract was changed to John H. Stroger Jr. Hospital of Cook County.

2. I made sure the term "comorbidities" are consistent throughout the manuscript (instead of co-morbidities).

3. I corrected a misspelled S. aureus on the last line on page 10.

The 2 abstracts match exactly now.

Thank you.